UCLA

UCLA Previously Published Works

Title

Mediating roles of preterm birth and restricted fetal growth in the relationship between maternal education and infant mortality: A Danish population-based cohort study

Permalink

https://escholarship.org/uc/item/8jn6b9kt

Journal

PLOS Medicine, 16(6)

ISSN

1549-1277

Authors

Yu, Yongfu

Liew, Zeyan

Wang, Aolin

et al.

Publication Date

2019

DOI

10.1371/journal.pmed.1002831

Peer reviewed





OPEN ACCESS

Citation: Yu Y, Liew Z, Wang A, Arah OA, Li J, Olsen J, et al. (2019) Mediating roles of preterm birth and restricted fetal growth in the relationship between maternal education and infant mortality: A Danish population-based cohort study. PLoS Med 16(6): e1002831. https://doi.org/10.1371/journal. pmed.1002831

Academic Editor: Lars Åke Persson, London School of Hygiene and Tropical Medicine, UNITED KINGDOM

Received: October 11, 2018

Accepted: May 20, 2019

Published: June 14, 2019

Copyright: © 2019 Yu et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All data is stored at the secure platform of Denmark Statistics, which is the central authority on Danish statistics with the mission to collect, compile and publish statistics on the Danish society. Due to restrictions related to Danish law and protecting patient privacy, the combined set of data as used in this study can only be made available through a trusted third party, Statistics Denmark (https://www.dst.dk/en/kontakt). This state organisation holds the data

RESEARCH ARTICLE

Mediating roles of preterm birth and restricted fetal growth in the relationship between maternal education and infant mortality: A Danish population-based cohort study

Yongfu Yun 1,2,3,4, Zeyan Liew 5,6, Aolin Wang 7,8, Onyebuchi A. Arah 2,9, Jialiang Li 10,11, Jørn Olsen 5,6, Sven Cnattingius 2, Guoyou Qin 3, Carsten Obel 4, Bo Fun 3,4,15‡*, Jiong Li 1,16‡*

- 1 Department of Clinical Epidemiology, Aarhus University, Aarhus, Denmark, 2 Department of Epidemiology, Fielding School of Public Health, University of California, Los Angeles, Los Angeles, California, United States of America, 3 Great Ormond Street Institute of Child Health, University College London, London, United Kingdom, 4 Administrative Data Research Centre for England, University College London, London, United Kingdom, 5 Department of Environmental Health Sciences, Yale School of Public Health, New Haven, Connecticut, United States of America, 6 Yale Center for Perinatal, Pediatric, and Environmental Epidemiology, Yale School of Public Health, New Haven, Connecticut, United States of America, 7 Program on Reproductive Health and the Environment, Department of Obstetrics, Gynecology and Reproductive Sciences, University of California, San Francisco, San Francisco, California, United States of America, 8 Institute for Computational Health Sciences, University of California, San Francisco, San Francisco, California, United States of America, 9 Department of Statistics, College of Letters and Science, University of California, Los Angeles, Los Angeles, California, United States of America, 10 Duke-NUS Medical School, National University of Singapore, Singapore, 11 Department of Statistics & Applied Probability, National University of Singapore, Singapore, 12 Clinical Epidemiology Unit, Department of Medicine, Solna, Karolinska University Hospital, Karolinska Institute, Stockholm, Sweden, 13 Department of Biostatistics, School of Public Health, Key Laboratory of Public Health Safety, Fudan University, Shanghai, China, 14 Section for General Medical Practice, Department of Public Health, Aarhus University, Aarhus, Denmark, 15 School of Data Science, Fudan University, Shanghai, China, 16 Ministry of Education-Shanghai Key Laboratory of Children's Environmental Health, Xinhua Hospital, Shanghai Jiao Tong University School of Medicine, Shanghai, China
- ‡ These authors are joint senior authors on this work.
- * fu@fudan.edu.cn (BF); jl@clin.au.dk (JL)

Abstract

Background

Socioeconomic disparities in infant mortality have persisted for decades in high-income countries and may have become stronger in some populations. Therefore, new understandings of the mechanisms that underlie socioeconomic differences in infant deaths are essential for creating and implementing health initiatives to reduce these deaths. We aimed to explore whether and the extent to which preterm birth (PTB) and small for gestational age (SGA) at birth mediate the association between maternal education and infant mortality.

Methods and findings

We developed a population-based cohort study to include all 1,994,618 live singletons born in Denmark in 1981–2015. Infants were followed from birth until death, emigration, or the day before the first birthday, whichever came first. Maternal education at childbirth was



used for this study. University-based Danish scientific organisations can be authorized to work with data within Statistics Denmark and such organisation can provide access to individual scientists inside and outside of Denmark.

Researchers can apply for access to these data when the request is approved by the Danish Data Protection Agency: https://www.datatilsynet.dk, the email address for the Danish Data Protection

Agency is: dt@datatilsynet.dk. Requests for data may be sent to Statistics Denmark: http://www.dst.dk/en/OmDS/organisation/TelefonbogOrg.aspx? kontor=13&tlfbogsort=sektion or the Danish Data Protection Agency: https://www.datatilsynet.dk.

Funding: YY is supported by unrestricted grant from the Lundbeck Foundation (R232-2016-2462 and R265-2017-4069, https://www. lundbeckfonden.com/en/), from the Danish Council for Independent Research (Project no. DFF-6110-00019, https://ufm.dk/en), and the 2016 Karen Elise Jensens Fond (http://www.kejfond.dk/). AW is supported by grant from the U.S. National Institute of Environmental Health Science (R01ES027051, https://www.niehs.nih.gov/index.cfm), and grant from the U.S. National Library of Medicine (K01LM012381, https://www.nlm.nih.gov/). JL is supported by he Nordic Cancer Union (grants 176673. 186200. and R217-A13234-18-S65. http://www.ncu.nu/), National Natural Science Foundation of China (81530086, http://www.nsfc. gov.cn/english/site_1/index.html), and Novo Nordisk Fonden (NNF180C0052029, https:// novonordiskfonden.dk/en/). OAA was supported by NIH grant from the National Center for Advancing Translational Science (NCATS, R01HD072296, https://ncats.nih.gov/), grant from the Eunice Kennedy Shriver National Institute of Child Health and Human Development (NICHD, R01HD072296, https://www.nichd.nih.gov/), and the facilities and resources provided by the California Center for Population Research at UCLA (CCPR), which receives core support from the Eunice Kennedy Shriver National Institute of Child Health and Human Development (NICHD, R24HD041022, https://www.nichd.nih.gov/). BF was supported by the UK Medical Research Council research grant (MR/M025152/1) and is supported by a start-up research grant at Fudan University. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

Abbreviations: CDE, controlled direct effect; IPMW, inverse-probability-of-mediators weight; IPTW, inverse-probability-of-treatment weight;

categorized as low, medium, or high. An inverse probability weighting of marginal structural models was used to estimate the controlled direct effect (CDE) of maternal education on offspring infant mortality, further split into neonatal (0-27 days) and postneonatal (28-364 days) deaths, and portion eliminated (PE) by eliminating mediation by PTB and SGA. The proportion eliminated by eliminating mediation by PTB and SGA was reported if the mortality rate ratios (MRRs) of CDE and PE were in the same direction. The MRRs between maternal education and infant mortality were 1.63 (95% CI 1.48-1.80, P < 0.001) and 1.19 (95% CI 1.08–1.31, P < 0.001) for low and medium versus high education, respectively. The estimated proportions of these total associations eliminated by reducing PTB and SGA together were 55% (MRR_{PE} = 1.27, 95% CI 1.15-1.40, P < 0.001) for low and 60% (MRR_{PE} = 1.11, 95% CI 1.01–1.22, P = 0.037) for medium versus high education. The proportions eliminated by eliminating PTB and SGA separately were, respectively, 46% and 11% for low education (versus high education) and 48% and 13% for medium education (versus high education). PTB and SGA together contributed more to the association of maternal educational disparities with neonatal mortality (proportion eliminated: 75%-81%) than with postneonatal mortality (proportion eliminated: 21%-23%). Limitations of the study include the untestable assumption of no unmeasured confounders for the causal mediation analysis, and the limited generalizability of the findings to other countries with varying disparities in access and quality of perinatal healthcare.

Conclusions

PTB and SGA may play substantial roles in the relationship between low maternal education and infant mortality, especially for neonatal mortality. The mediating role of PTB appeared to be much stronger than that of SGA. Public health strategies aimed at reducing neonatal mortality in high-income countries may need to address socially related prenatal risk factors of PTB and impaired fetal growth. The substantial association of maternal education with postneonatal mortality not accounted for by PTB or SGA could reflect unaddressed educational disparities in infant care or other factors.

Author summary

Why was this study done?

- Infant mortality in high-income countries has decreased in recent decades.
- However, socioeconomic inequality in infant mortality remains and may have become stronger in some populations.
- This study provides in-depth knowledge on the underlying pathways from maternal socioeconomic inequality to infant mortality, which is important for developing preventive strategies to reduce potentially preventable deaths.



LBW, low birth weight; MRR, mortality rate ratio; MSM, marginal structural model; PE, portion eliminated; PTB, preterm birth; SGA, small for gestational age; TE, total effect.

What did the researchers do and find?

- We conducted a national population-based cohort study that included all 1.99 million live singletons born in Denmark in 1981–2015.
- Using modern modeling methods for causal mediation analysis, we found that preterm birth and restricted fetal growth may mediate the association between low maternal education and infant mortality.
- The mediated effects through preterm birth and restricted fetal growth were substantial for neonatal mortality but not for postneonatal mortality. The effect through preterm birth seemed to be greater than that through restricted fetal growth.

What do these findings mean?

- Public health strategies could aim to reduce educational differences in the death of newborns in high-income countries by addressing socially related risk factors for preterm birth and impaired fetal growth, such as the accessibility and quality of maternity and perinatal care, or maternal lifestyle factors.
- The substantial direct impact of low maternal education on postneonatal death not
 explained by preterm birth or restricted fetal growth could be due to unaddressed educational inequality in infant care or other factors such as housing conditions, the quality
 of medical care, or the use of specialized medical care.

Introduction

Infant mortality in high-income countries has decreased over the last decades [1]. Nevertheless, socioeconomic disparities in infant mortality persist [2-5] and may even have become stronger in some populations [6,7]. A new understanding of the mechanisms that underlie socioeconomic differences in infant deaths is essential to guide health initiatives to reduce potentially preventable deaths.

Preterm birth (PTB), small for gestational age (SGA), and low birth weight (LBW) are not only main risk indicators for infant morbidity and mortality, particularly during the neonatal period, but are also associated with socioeconomic disadvantage [4,8–14]. The primary causes of LBW are PTB and fetal growth restriction [12]. It is possible that the association between socioeconomic disadvantage and infant mortality is mediated through PTB and SGA as a proxy of fetal growth restriction [12,15]. PTB and fetal growth restriction may reflect different mechanisms, and the risk factors for PTB and fetal growth restriction are different [16,17]. The risk of infant mortality is higher among preterm-born infants than SGA infants [12,18]. However, there is a lack of research addressing the possible mediating role of PTB and SGA in the pathway linking maternal socioeconomic disadvantage to an increased risk of infant mortality. Causal mediation analysis may help advance our understanding of when and how socioeconomic disadvantage has a large impact on infant mortality. If the mediating pathway through PTB and SGA plays a major role, policies should focus on the health of women before and during pregnancy to reduce the risk of PTB and SGA. Otherwise, interventions targeting other factors should be explored.



Specifically, using data from Danish registers, we aimed to use causal mediation analysis to examine whether and the extent to which PTB and SGA mediate the association of maternal education with infant mortality. Analyses were stratified by the time (neonatal or postneonatal period) and cause (disease or external cause) of death.

Methods

Ethics statement

The study was approved by the Data Protection Agency and Research Ethics Committee of the Central Region in Denmark. By Danish law, no informed consent is required for a register-based study using anonymized data. The prespecified analysis plan is given in S1 Text.

Study design and participants

The unique personal identification number in Denmark allows accurate linkage of personal data across national registers. The Danish Civil Registration System was linked to the Danish National Patient Register, the Danish Medical Birth Register, the Danish Register of Causes of Death, and the Danish Integrated Database for Labor Market Research [19]. There were 2,096,320 live singleton births registered in Denmark in 1981–2015. We excluded infants with birthweight less than 500 grams or gestational age at birth less than 22 weeks, as criteria for registration of live births and stillbirth vary internationally [20]. We also excluded infants with missing information on maternal education. Follow-up started at birth and ended at death, emigration, or the day before the first birthday, whichever came first.

Maternal education

The Danish Integrated Database for Labor Market Research [19] provided the information on maternal education, which was measured as the highest level of education attained at child-birth and categorized as low (primary and lower secondary education), medium (upper secondary education or academy profession degree), or high (university education at bachelor's degree level or higher).

Outcomes

The outcome of interest was all-cause infant mortality (0–364 days), which was divided into neonatal mortality (0–27 days) and postneonatal mortality (28–364 days). We also investigated mortality by cause of death (death due to disease or medical condition, or external cause), as well as death due to certain conditions originating in the perinatal period [ICD-8 codes 760–779 and ICD-10 P00–P96] and congenital malformations [ICD-8 740–759 and ICD-10 Q00–Q99], according to the European Shortlist for Causes of Death [21].

Potential mediators

Birth weight and gestational age were extracted from the Danish Medical Birth Register [22]. Gestational age was estimated by the date of last menstrual period and, for all pregnancies since 1995, was adjusted, if necessary, by ultrasonography. The mediators were dichotomized as PTB (1 if gestational age at birth < 37 weeks, 0 otherwise) and SGA (1 if birthweight below the 10th percentile for infants of the same gestational age, sex, and birth year, 0 otherwise) in the main analyses. We also used finer categorizations of PTB (<28, 28–31, 32–36, or 37 + weeks) and SGA (birthweight below the 3rd, between the 3rd and 10th, or above the 10th percentile for infants of the same gestational age, sex, and birth year).



Covariates

Potential confounders (covariates) included maternal age at birth (<20, 20–24, 25–29, 30–34, \ge 35 years), parity (1, 2, 3, \ge 4), maternal smoking at delivery (yes, no), maternal cohabitation at delivery (single, cohabitation), maternal residence at delivery (Copenhagen, city with \ge 100,000 inhabitants, small town/other), diagnosis of congenital malformation of the infant (yes, no), infant sex (male, female), and birth year of the infant (in 5-year intervals). Maternal smoking and congenital malformation and sex of the child were used to adjust for the confounding between the mediators and the outcome when estimating controlled direct effect (CDE); we preserved their potential mediating roles between the exposure and the outcome (in the statistical analysis).

Statistical analysis

The approach for causal mediation analysis was based on a counterfactual framework whereby the total effect (TE) can be decomposed into controlled direct effect (CDE) and portion eliminated (PE) [23-25]. The CDE captured the influence of maternal education on infant (neonatal and postneonatal) mortality if the link between maternal education and the mediator (PTB or SGA) was prevented or removed hypothetically. This simulated a scenario wherein the sample distributions of the mediator were no longer dependent on maternal education. PE, the difference between TE and CDE, measured the portion of the TE of maternal education that would be eliminated by eliminating the mediator. TE and CDE were estimated using inverseprobability-weighted marginal structural models (MSMs) [26]. For the MSM for the TE, we used weighted regressions of infant mortality on maternal education. The weight, which is called the inverse-probability-of-treatment weight (IPTW), was estimated for each mother in the sample as the ratio of (i) the estimated marginal probability of the mother's actual educational attainment to (ii) the estimated probability of each mother's actual educational attainment conditional on their aforementioned covariates (excluding maternal smoking, offspring sex, congenital malformation of offspring, PTB, and SGA). The IPTW simulates the scenario wherein these covariates, which could be confounders, are no longer associated with maternal education, thus eliminating any confounding by these covariates. To estimate the CDE, the corresponding MSM used a product of the IPTW for maternal education and an additional inverse-probability-of-mediators weight (IPMW). The IPMW was estimated for each infant in the sample as the ratio of (i) the estimated marginal probability of the infant's actual PTB and SGA to (ii) the estimated probability of each infant's actual PTB and SGA conditional on their aforementioned covariates. The IPMW simulates the scenario wherein the mediators (PTB and SGA) are no longer associated with maternal education, thus eliminating any mediation by the mediators. The PE was subsequently estimated from the model for TE offsetting the estimated CDE. We considered possible exposure-mediator and mediator-mediator interactions. As the mediator-mediator interactions in causal mediation analysis with multiple mediators were null, the final models included only exposure-mediator interactions. We estimated mortality rate ratios (MRRs) with their 95% confidence intervals (CIs) based on robust variance estimation. The proportion of the TE eliminated through the 2 mediators, i.e., the proportion eliminated ([MRR_{TE}-MRR_{CDE}]/[MRR_{TE}-1]), was reported if the MRRs of CDE and PE were in the same direction [23,27]. We first assessed the mediating role of PTB and SGA separately, i.e., one mediator at a time. Then we analyzed PTB and SGA together as a joint mediator, i.e., not separating their individual contributions. We also examined the mediating roles of PTB in non-SGA infants as well as the mediating role of SGA in term-born infants. The mediation analyses were performed according to birth year (1981-2015 in 5-year intervals). We used the missing-value indicator method to deal with missing values, such that



missing values were treated as a separate category. A detailed description of the inverse probability weighting approach for causal mediation analysis is given in <u>\$2 Text</u>. We also performed additional mediation analysis using a traditional approach ("with and without mediator") [28].

Sensitivity analysis

We performed a sensitivity analysis to assess and adjust for violations of the uncontrolled confounding assumption [23,29]. Specifically, we considered a binary unmeasured confounding variable U indicating a common cause of PTB, SGA, and infant mortality (e.g., maternal alcohol use, maternal body mass index, prenatal care, maternal psychological stress in adolescence, or maternal school attendance [30,31]). We assumed that among infants with normal gestational age and normal birth weight for gestational age, the prevalence of U was 20% for the low maternal education group, 30% for the medium maternal education group, and 40% for the high maternal education group. We also considered a simplified assumption that the prevalence of U was the same among different maternal education groups. We evaluated the impact of unmeasured mediator–outcome confounding in 2 settings: (i) moderate confounding, where we considered if U increased the likelihood of infant mortality by a factor of 1.5, and (ii) strong confounding, where we considered if U increased the likelihood of infant mortality by a factor of 2.5 [30]. We also performed analysis with an additional adjustment for maternal country of origin. All analyses were conducted using SAS 9.4 (SAS Institute, Cary, NC, US) and Stata 13 (StataCorp, College Station, TX, US).

Results

The study included 1,994,618 infants. Excluded infants (N = 101,702,4.85%) had a higher infant mortality rate (1.46%) than included infants (0.43%) and tended to have a younger mother. Overall, infant mortality was 4.3 per 1,000 births (8,563 died), and 61.86% of deaths occurred in the neonatal period. Compared with infants of mothers with medium or high education, infants of mothers with low education were more likely to have increased risk of death, to have PTB, to be SGA, and to be born of mothers of younger childbearing age, with higher parity, who lived alone, and who were from small towns (Table 1).

Regarding the total association of maternal education with mortality, we observed that MRRs decreased with increasing education level (Table 2; S1 Fig). The MRR_{TE} of association with low maternal education (versus high) was 1.63 (95% CI 1.48–1.80, P < 0.001) for infant mortality (neonatal and postneonatal), 1.57 (95% CI 1.38–1.78, P < 0.001) for neonatal mortality, and 1.75 (95% CI 1.49–2.04, P < 0.001) for postneonatal mortality. We found that PTB and SGA were associated with both maternal education and infant mortality, and the association between PTB and infant mortality was much stronger than the association between SGA and infant mortality (S1 Table). Analyses stratified by birth year found a stronger association between low maternal education and infant mortality in recent years (S2 Table).

Mediation analysis including PTB and SGA together showed that PTB and SGA played an important role in explaining the link between maternal education and all-cause infant mortality (Table 2; S1 Fig). Compared with high maternal education, the estimated proportion of the total association of education with infant death that could be reduced by "eliminating" the mediating role of PTB and SGA was 55% (MRR_{PE} = 1.27 [95% CI 1.15–1.40, P < 0.001]) for low education and 60% (MRR_{PE} = 1.11 [95% CI 1.01–1.22, P = 0.037]) for medium education. Regarding neonatal mortality, excess deaths were mainly due to the pathway involving PTB and SGA (low versus high education: proportion eliminated = 75%, MRR_{PE} = 1.37 [95% CI 1.21–1.56, P < 0.001]; medium versus high education: proportion eliminated = 81%, MRR_{PE} = 1.14 [95% CI 1.01–1.28, P = 0.032]). During the postneonatal period, other pathways, rather



Table 1. Baseline characteristics for mother and child across maternal education at birth.

Characteristic	Maternal education g	Total			
	Low	Medium	High		
Number of infants	520,642 (26.1)	893,388 (44.8)	580,588 (29.1)	1,994,618	
Infant deaths	3,398 (0.7)	3,452 (0.4)	1,713 (0.3)	8,563 (0.4)	
Preterm birth					
<37 weeks	27,855 (5.4)	37,354 (4.2)	19,761 (3.4)	84,970 (4.3)	
37+ weeks	492,787 (94.7)	856,034 (95.8)	560,827 (96.6)	1,909,648 (95.7)	
Small for gestational age					
Yes	64,390 (12.4)	82,894 (9.3)	44,961 (7.7)	192,245 (9.6)	
No	456,252 (87.6)	810,494 (90.7)	535,627 (92.3)	1,802,373 (90.4)	
Sex					
Boy	267,352 (51.4)	458,152 (51.3)	298,551 (51.4)	1,024,055 (51.3)	
Girl	253,290 (48.7)	435,236 (48.7)	282,037 (48.6)	970,563 (48.7)	
Congenital malformation					
No	508,165 (97.6)	873,171 (97.7)	567,462 (97.7)	1,948,798 (97.7)	
Yes	12,477 (2.4)	20,217 (2.3)	13,126 (2.3)	45,820 (2.3)	
Maternal age at birth (years)					
<20	38,843 (7.5)	3,240 (0.4)	93 (0.0)	42,176 (2.1)	
20–24	169,911 (32.6)	153,684 (17.2)	10,757 (1.9)	334,352 (16.8)	
25–29	169,311 (32.5)	370,095 (41.4)	191,484 (33.0)	730,890 (36.6)	
30–34	96,893 (18.6)	260,190 (29.1)	254,203 (43.8)	611,286 (30.7)	
35+	45,684 (8.8)	106,179 (11.9)	124,051 (21.4)	275,914 (13.8)	
Parity					
1	220,198 (42.3)	419,604 (47.0)	252,475 (43.5)	892,277 (44.7)	
2	178,464 (34.3)	334,980 (37.5)	230,079 (39.6)	743,523 (37.3)	
3	81,080 (15.6)	107,746 (12.1)	81,029 (14.0)	269,855 (13.5)	
≥4	40,900 (7.9)	31,058 (3.5)	17,005 (2.9)	88,963 (4.5)	
Maternal residence at birth					
Copenhagen	47,813 (9.2)	80,670 (9.0)	93,635 (16.1)	222,118 (11.1)	
City with \geq 100,000 inhabitants	58,696 (11.3)	106,049 (11.9)	94,289 (16.2)	259,034 (13.0)	
Small town/other	414,133 (79.5)	706,669 (79.1)	392,664 (67.6)	1,513,466 (75.9)	
Maternal cohabitation at birth					
Single	291,133 (55.9)	408,679 (45.7)	229,668 (39.6)	929,480 (46.6)	
Cohabitation	229,509 (44.1)	484,709 (54.3)	350,920 (60.4)	1,065,138 (53.4)	
Maternal smoking at birth ^a					
No	175,642 (56.4)	529,843 (77.3)	426,139 (89.6)	1,131,624 (76.9)	
Yes	122,621 (39.4)	131,263 (19.1)	36,115 (7.6)	289,999 (19.7)	
Unknown	13,054 (4.2)	24,655 (3.6)	13,165 (2.8)	50,874 (3.5)	
Birth year					
1981–1985	106,293 (20.4)	87,920 (9.8)	47,592 (8.2)	241,805 (12.1)	
1986–1990	103,032 (19.8)	119,707 (13.4)	57,577 (9.9)	280,316 (14.1)	
1991–1995	93,659 (18.0)	157,209 (17.6)	68,389 (11.8)	319,257 (16.0)	
1996–2000	72,893 (14.0)	156,065 (17.5)	77,183 (13.3)	306,141 (15.4)	
2001–2005	58,160 (11.2)	145,453 (16.3)	97,506 (16.8)	301,119 (15.1)	
2006–2010	49,143 (9.4)	127,187 (14.2)	117,568 (20.3)	293,898 (14.7)	
2011–2015	37,462 (7.2)	99,847 (11.2)	114,773 (19.8)	252,082 (12.6)	

Data are expressed as frequency (percentage). Percentages have been rounded and may not total 100.

^aMaternal smoking was available from 1991 to 2015.

https://doi.org/10.1371/journal.pmed.1002831.t001



Table 2. The contribution of PTB and SGA in explaining the association between maternal education and infant mortality.

Mediator	Period	Education	Number of deaths	Rate/10 ² pys	MRR _{TE}	P value	MRR _{CDE}	P value	MRR _{PE}	P value	Proportion eliminated
PTB	Infant (<1 year)	Low	3,398	6.58	1.63 (1.48- 1.80)	0.000	1.34 (1.21- 1.49)	0.000	1.22 (1.10- 1.34)	0.000	46%
		Medium	3,452	3.89	1.19 (1.08– 1.31)	0.000	1.10 (0.99– 1.22)	0.070	1.08 (0.99– 1.19)	0.094	48%
		High	1,713	2.97	1.00 (reference)						
	Neonatal (0–27 days)	Low	1,944	50.67	1.57 (1.38– 1.78)	0.000	1.20 (1.05– 1.37)	0.007	1.30 (1.15– 1.48)	0.000	64%
		Medium	2,268	34.42	1.18 (1.05– 1.33)	0.006	1.07 (0.94– 1.21)	0.330	1.11 (0.99– 1.25)	0.087	64%
		High	1,158	27.03	1.00 (reference)						
	Postneonatal (28–364	Low	1,454	3.04	1.75 (1.49– 2.04)	0.000	1.62 (1.36– 1.92)	0.000	1.08 (0.93– 1.26)	0.328	17%
	days)	Medium	1,184	1.44	1.21 (1.04– 1.41)	0.015	1.17 (0.98– 1.38)	0.075	1.04 (0.89– 1.21)	0.650	20%
		High	555	1.04	1.00 (reference)						
SGA	Infant (<1 year)	Low	3,398	6.58	1.63 (1.48– 1.80)	0.000	1.56 (1.42– 1.72)	0.000	1.05 (0.95– 1.15)	0.377	11%
		Medium	3,452	3.89	1.19 (1.08– 1.31)	0.000	1.17 (1.06– 1.28)	0.001	1.02 (0.93– 1.12)	0.662	13%
		High	1,713	2.97	1.00 (reference)						
	Neonatal (0–27 days)	Low	1,944	50.67	1.57 (1.38– 1.78)	0.000	1.50 (1.32- 1.70)	0.000	1.05 (0.92– 1.19)	0.478	12%
		Medium	2,268	34.42	1.18 (1.05– 1.33)	0.006	1.15 (1.02– 1.29)	0.018	1.03 (0.91– 1.16)	0.661	17%
		High	1,158	27.03	1.00 (reference)						
	Postneonatal (28–364	Low	1,454	3.04	1.75 (1.49– 2.04)	0.000	1.68 (1.44– 1.96)	0.000	1.04 (0.89– 1.21)	0.648	8%
	days)	Medium	1,184	1.44	1.21 (1.04– 1.41)	0.015	1.20 (1.03- 1.39)	0.020	1.01 (0.87- 1.18)	0.902	6%
		High	555	1.04	1.00 (reference)						
PTB and SGA	Infant (<1 year)	Low	3,398	6.58	1.63 (1.48– 1.80)	0.000	1.28 (1.16- 1.42)	0.000	1.27 (1.15- 1.40)	0.000	55%
		Medium	3,452	3.89	1.19 (1.08– 1.31)	0.000	1.08 (0.97– 1.19)	0.141	1.11 (1.01- 1.22)	0.037	60%
		High	1,713	2.97	1.00 (reference)						
	Neonatal (0-27 days)	Low	1,944	50.67	1.57 (1.38– 1.78)	0.000	1.14 (1.00- 1.30)	0.053	1.37 (1.21- 1.56)	0.000	75%
		Medium	2,268	34.42	1.18 (1.05– 1.33)	0.006	1.03 (0.91– 1.17)	0.553	1.14 (1.01– 1.28)	0.032	81%
		High	1,158	27.03	1.00 (reference)						
	Postneonatal (28–364 days)	Low	1,454	3.04	1.75 (1.49– 2.04)	0.000	1.58 (1.33– 1.87)	0.000	1.11 (0.95- 1.29)	0.172	23%
		Medium	1,184	1.44	1.21 (1.04– 1.41)	0.015	1.17 (0.99– 1.38)	0.076	1.04 (0.89– 1.21)	0.614	21%
		High	555	1.04	1.00 (reference)						

Proportion eliminated = $(MRR_{TE} - MRR_{CDE})/(MRR_{TE} - 1)$; proportion eliminated is only presented if the MRRs of CDE and PE were in the same direction. CDE, controlled direct effect; MRR, mortality rate ratio; PE, portion eliminated; PTB, preterm birth; pys, person-years; SGA, small for gestational age; TE, total effect.

https://doi.org/10.1371/journal.pmed.1002831.t002



Table 3. The contribution of PTB and SGA in explaining the association between maternal education and infant mortality using a finer categorization of PTB and SGA.

Mediator	Period	Education	MRR _{TE}	P value	MRR _{CDE}	P value	MRR _{PE}	P value	Proportion eliminated
PTB	Infant	Low	1.63 (1.48-1.80)	0.000	1.21 (1.09-1.36)	0.001	1.34 (1.22-1.48)	0.000	66%
	(<1 year)	Medium	1.19 (1.08–1.31)	0.000	1.05 (0.95-1.17)	0.338	1.13 (1.03-1.24)	0.011	72%
		High	1.00 (reference)						
	Neonatal (0-27 days)	Low	1.57 (1.38–1.78)	0.000	1.06 (0.92-1.21)	0.441	1.48 (1.31-1.68)	0.000	90%
		Medium	1.18 (1.05–1.33)	0.006	1.02 (0.89-1.16)	0.800	1.16 (1.03-1.31)	0.013	90%
		High	1.00 (reference)						
	Postneonatal	Low	1.75 (1.49–2.04)	0.000	1.55 (1.30-1.85)	0.000	1.12 (0.96-1.31)	0.137	26%
	(28-364 days)	Medium	1.21 (1.04–1.41)	0.015	1.14 (0.95–1.36)	0.148	1.06 (0.91-1.24)	0.446	33%
		High	1.00 (reference)						
SGA	Infant	Low	1.63 (1.48-1.80)	0.000	1.55 (1.40-1.70)	0.000	1.06 (0.96-1.16)	0.282	14%
	(<1 year)	Medium	1.19 (1.08–1.31)	0.000	1.16 (1.06–1.27)	0.002	1.03 (0.94-1.13)	0.569	17%
		High	1.00 (reference)						
	Neonatal (0–27 days)	Low	1.57 (1.38–1.78)	0.000	1.48 (1.30–1.67)	0.000	1.06 (0.93-1.20)	0.361	16%
		Medium	1.18 (1.05–1.33)	0.006	1.14 (1.02-1.28)	0.026	1.04 (0.92-1.17)	0.558	23%
		High	1.00 (reference)						
	Postneonatal (28–364 days)	Low	1.75 (1.49–2.04)	0.000	1.68 (1.44-1.96)	0.000	1.04 (0.89-1.21)	0.617	9%
		Medium	1.21 (1.04–1.41)	0.015	1.20 (1.03-1.39)	0.021	1.01 (0.87-1.18)	0.882	7%
		High	1.00 (reference)						
PTB and SGA	Infant (<1 year)	Low	1.63 (1.48-1.80)	0.000	1.16 (1.04-1.30)	0.007	1.40 (1.27-1.55)	0.000	74%
		Medium	1.19 (1.08–1.31)	0.000	1.03 (0.93-1.15)	0.527	1.15 (1.05–1.26)	0.003	82%
		High	1.00 (reference)						
	Neonatal (0–27 days)	Low	1.57 (1.38–1.78)	0.000	1.00 (0.88-1.15)	0.975	1.56 (1.38–1.77)	0.000	100%
		Medium	1.18 (1.05–1.33)	0.006	0.99 (0.87-1.13)	0.877	1.19 (1.06–1.34)	0.003	100%
		High	1.00 (reference)						
	Postneonatal (28–364 days)	Low	1.75 (1.49–2.04)	0.000	1.54 (1.30-1.83)	0.000	1.13 (0.97-1.32)	0.114	28%
		Medium	1.21 (1.04–1.41)	0.015	1.15 (0.97–1.37)	0.100	1.05 (0.90-1.22)	0.545	27%
		High	1.00 (reference)						

Proportion eliminated = $(MRR_{TE} - MRR_{CDE})/(MRR_{TE} - 1)$; proportion eliminated is only presented if the MRRs of CDE and PE were in the same direction. A finer categorization of PTB (<28, 28-31, 32-36, 37+ weeks) and SGA (birthweight below the 3rd, between the 3rd and 10th, or above the 10th percentile for infants of the same gestational age, sex, and birth year).

CDE, controlled direct effect; MRR, mortality rate ratio; PE, portion eliminated; PTB, preterm birth; SGA, small for gestational age; TE, total effect.

https://doi.org/10.1371/journal.pmed.1002831.t003

than these mediators, may play a major role in explaining the increased rate of deaths (low versus high education: proportion eliminated = 23%, MRR_{PE} = 1.11 [95% CI 0.95–1.29, P = 0.172]; medium versus high education: proportion eliminated = 21%, MRR_{PE} = 1.04 [95% CI 0.89–1.21, P = 0.614]). Regarding the mediation analyses including one mediator at a time (Table 2; S1 Fig), the mediating role of PTB (low versus high education: proportion eliminated = 46%, MRR_{PE} = 1.22 [95% CI 1.10–1.34, P < 0.001]; medium versus high education: proportion eliminated = 48%, MRR_{PE} = 1.08 [95% CI 0.99–1.19, P = 0.094]) in the association between maternal education and infant mortality was much stronger than that of SGA (low versus high education: proportion eliminated = 11%, MRR_{PE} = 1.05 [95% CI 0.95–1.15, P = 0.377]; medium versus high education: proportion eliminated = 13%, MRR_{PE} = 1.02 [95% CI 0.93–1.12, P = 0.662]). Similar patterns were also found for neonatal mortality and postneonatal mortality analyzed separately. A finer categorization of the mediators yielded a stronger mediating impact by PTB and SGA (Table 3).



Mediation analyses stratified by birth year yielded results similar to those from the main analyses, although the estimated proportion of the total association between maternal education and infant death reduced by eliminating the mediating roles of PTB and SGA seemed to decrease in recent years (S2 and S3 Tables). Analyses restricted to non-SGA infants on the mediating role of PTB found a similar pattern as the analyses considering PTB and SGA together (S4 Table). We found that SGA weakly mediated the association between maternal education and infant mortality among term-born infants (S5 Table).

Deaths due to diseases accounted for 96.8% (8,292 died) of all infant deaths, and the pattern observed for infant deaths due to diseases was similar to that of all-cause mortality (Table 4). Similar patterns were also found for deaths due to certain conditions originating in the perinatal period and congenital malformations (S6 and S7 Tables). Especially for infant deaths due to certain conditions originating in the perinatal period, a substantial mediating impact through PTB and SGA together was found (low versus high education: proportion mediated = 76%, MRR_{PE} = 1.49 [95% CI 1.26–1.76, P < 0.001]). A much smaller portion of deaths due to external causes was found to be mediated by PTB and SGA (S8 Table).

In addition, we performed sensitivity analyses to assess the potential impact of unmeasured confounding. Under the simplified assumption that the prevalence of U was the same in each maternal education group, the CDE was unaffected. However, if the prevalence of U was assumed to differ between maternal education groups, the adjusted CDE was higher than the original CDE. Even if the unmeasured confounder was strong enough to increase the likelihood of infant death by 2.5-fold, we still observed the mediating impact of PTB on neonatal mortality (S9 Table). An additional adjustment for maternal country of origin (S10 Table) did not essentially change the results. The results from the traditional approach with and without mediators for mediation analysis differed from those of our MSM approach (S11 Table).

Discussion

We found that low maternal education was associated with a higher risk of infant mortality in Denmark, and this association was partly mediated by PTB and SGA. The mediatory role of PTB and SGA was substantial in neonatal mortality, while a large direct impact of low maternal education on postneonatal mortality was observed. The mediating effects through PTB were greater than those through SGA.

A number of studies [3,5,9,10,32] have investigated the association between maternal socioeconomic disadvantage and infant mortality, but few of them have attempted to distinguish the relative roles of LBW, PTB, and SGA in the association. A British study [15], using a traditional method to explore the association between social class and infant mortality ("with and without mediator"), reported that LBW was a strong risk factor in the neonatal period but did not seem to play an important role in the postneonatal period. The traditional regression approach, including a potential mediator as a covariate, is easily implemented and understood. However, the traditional mediation analysis method is prone to yielding a flawed conclusion due to exposure-mediator interaction, mediator-outcome confounding, and mediator-outcome confounding affected by the exposure (intermediate confounding) [23,28]. Especially in the context of longitudinal design and time-varying confounders, intermediate confounders may not be rare. In a traditional mediation approach, adjustment for the mediator might lead to a spurious association between the intermediate confounder and the exposure, where the intermediate confounder becomes the confounder between the exposure and the outcome. Adjustment for intermediate confounders is required to prevent such bias, known as colliderstratification bias [23]. However, adjustment for intermediate confounders in a traditional mediation model could block part of the effect of exposure on outcome through the



Table 4. The combined contribution of PTB and SGA in explaining the association between maternal education and infant mortality due to diseases.

Mediator	Period	Education	Number of deaths	Rate/10 ² pys	MRR _{TE}	P value	MRR _{CDE}	P value	MRR _{PE}	P value	Proportion eliminated
РТВ	Infant (<1 year)	Low	3,288	6.37	1.62 (1.47– 1.79)	0.000	1.33 (1.19– 1.48)	0.000	1.22 (1.11– 1.35)	0.000	47%
		Medium	3,340	3.76	1.18 (1.07- 1.30)	0.001	1.09 (0.98– 1.21)	0.108	1.08 (0.99– 1.19)	0.095	51%
		High	1,664	2.88	1.00 (reference)						
	Neonatal (0–27 days)	Low	1,924	50.15	1.57 (1.38– 1.78)	0.000	1.20 (1.05– 1.37)	0.009	1.31 (1.15– 1.48)	0.000	65%
		Medium	2,231	33.86	1.17 (1.04– 1.32)	0.009	1.06 (0.93– 1.20)	0.392	1.11 (0.98– 1.25)	0.091	66%
		High	1,145	26.72	1.00 (reference)						
	Postneonatal (28–364	Low	1,364	2.86	1.73 (1.48– 2.03)	0.000	1.60 (1.34– 1.91)	0.000	1.08 (0.92– 1.27)	0.326	18%
	days)	Medium	1,109	1.35	1.20 (1.02- 1.41)	0.024	1.16 (0.97– 1.38)	0.109	1.04 (0.89– 1.22)	0.643	22%
		High	519	0.97	1.00 (reference)						
SGA	Infant (<1 year)	Low	3,288	6.37	1.62 (1.47– 1.79)	0.000	1.55 (1.41- 1.71)	0.000	1.05 (0.95– 1.16)	0.375	12%
		Medium	3,340	3.76	1.18 (1.07- 1.30)	0.001	1.16 (1.05– 1.27)	0.002	1.02 (0.93- 1.12)	0.655	14%
		High	1,664	2.88	1.00 (reference)						
	Neonatal (0–27 days)	Low	1,924	50.15	1.57 (1.38– 1.78)	0.000	1.50 (1.32– 1.70)	0.000	1.05 (0.92– 1.19)	0.477	12%
		Medium	2,231	33.86	1.17 (1.04– 1.32)	0.009	1.14 (1.01- 1.28)	0.028	1.03 (0.91– 1.16)	0.654	18%
		High	1,145	26.72	1.00 (reference)						
	Postneonatal (28–364	Low	1,364	2.86	1.73 (1.48– 2.03)	0.000	1.67 (1.42– 1.96)	0.000	1.04 (0.88– 1.22)	0.645	9%
	days)	Medium	1,109	1.35	1.20 (1.02- 1.41)	0.024	1.19 (1.02– 1.39)	0.031	1.01 (0.86- 1.18)	0.904	6%
		High	519	0.97	1.00 (reference)						
PTB and SGA	Infant (<1 year)	Low	3,288	6.37	1.62 (1.47– 1.79)	0.000	1.26 (1.14– 1.40)	0.000	1.28 (1.16– 1.42)	0.000	58%
		Medium	3,340	3.76	1.18 (1.07- 1.30)	0.001	1.07 (0.96– 1.18)	0.217	1.11 (1.01– 1.22)	0.035	63%
		High	1,664	2.88	1.00 (reference)						
	Neonatal (0-27 days)	Low	1,924	50.15	1.57 (1.38– 1.78)	0.000	1.13 (0.99– 1.29)	0.063	1.38 (1.22– 1.57)	0.000	76%
		Medium	2,231	33.86	1.17 (1.04– 1.32)	0.009	1.03 (0.91– 1.17)	0.662	1.14 (1.01– 1.28)	0.031	84%
		High	1,145	26.72	1.00 (reference)						
	Postneonatal (28–364	Low	1,364	2.86	1.73 (1.48- 2.03)	0.000	1.55 (1.30- 1.84)	0.000	1.12 (0.95– 1.31)	0.169	25%
	days)	Medium	1,109	1.35	1.20 (1.02- 1.41)	0.024	1.15 (0.97– 1.37)	0.110	1.04 (0.89– 1.22)	0.610	24%
		High	519	0.97	1.00 (reference)						

Deaths due to diseases: ICD-8 codes 000–799 and ICD-10 codes A00–R99. Proportion eliminated = $(MRR_{TE} - MRR_{CDE})/(MRR_{TE} - 1)$; proportion eliminated is only presented if the MRRs of CDE and PE were in the same direction.

CDE, controlled direct effect; MRR, mortality rate ratio; PE, portion eliminated; PTB, preterm birth; pys, person-years; SGA, small for gestational age; TE, total effect.

https://doi.org/10.1371/journal.pmed.1002831.t004



intermediate confounder, resulting in underestimation of the direct effect not through the mediator and overestimation of the mediating effect through the mediator [23,33]. Our findings also suggest that the traditional approach for mediation analysis is probably subject to bias (S11 Table). We used a counterfactual approach for causal mediation analysis that allows decomposing the TE into the CDE and the PE in the presence of exposure–mediator interaction [23]. Inverse-probability-weighted MSMs could address intermediate confounders and estimate the CDE, without blocking the pathway from the exposure to the outcome acting through the intermediate confounders (see S2 Text for details) [23,26]. Moreover, sensitivity analysis techniques based on a causal inference framework can be applied to evaluate the impact of unmeasured confounders [29].

Low socioeconomic status has been associated with PTB, SGA, and LBW [12–14,34,35]. Socioeconomic inequality may lead to differences in prenatal risk factors for PTB and SGA, including the accessibility/quality of maternity/perinatal care, maternal health behaviors, occupational situation, nutrition, and health outcomes [2,35–37]. These prenatal risk factors may have an influence on risks of PTB and fetal growth restriction, thereby leading to an increased risk of infant death [36]. Certain conditions originating in the perinatal period and congenital malformations are the 2 leading causes of infant death [38]. In mediation analyses of infant mortality due to certain conditions originating in the perinatal period and congenital malformations, we found similar mediating patterns as in the mediation analyses for overall infant mortality. PTB, SGA, and LBW are important indirect causes of neonatal deaths [12,36,39], and our results also showed that the mediating roles of PTB and SGA accounted for a large number of excess neonatal deaths related to low maternal education.

In accordance with previous studies, we observed that the total association of maternal education and infant mortality tended to be stronger in the postneonatal period than in the neonatal period [5,32]. Our finding that both the CDE and PE of maternal education on infant mortality differed between the neonatal and the postneonatal period may reflect separate causal pathways. The substantial impact of PTB and SGA on the association of maternal education with neonatal mortality due to diseases may reflect that the excess neonatal deaths are more influenced by prenatal and perinatal mechanisms, like PTB, placental impairment, and impaired fetal growth. Although neonatal mortality has been linked to the quality of obstetric and neonatal care [31], it is less likely that maternal disadvantage would significantly affect the care provided by the neonatal intensive care center in Denmark, where there is universal taxpaid health coverage and the utilization of these services is very high. Disorders related to short gestation and impaired fetal growth are important causes of neonatal death [8], and might partly explain why maternal education mainly acted through the mediating role of PTB and SGA.

In the postneonatal period, substantial direct impact of maternal education on offspring mortality was seen. Although Denmark has universal health coverage for all essential health services, education could reflect differences in both economic and non-economic factors. Higher levels of education are related to active acquisition of health-related knowledge and increased use of special health services. Education also indicates the ability to solve problems and the capacity to deal with stressors [40]. Infants with well-educated mothers are more likely to benefit from optimized use of health and social welfare services and to receive a higher quality of clinical care [31]. A previous study showed that children born to mothers with lower education used fewer general practitioner services and specialist services, and there were considerable differences in the use of telephone consultations with doctors according to maternal education [41]. Therefore, socioeconomic postnatal differences in the environmental and social circumstances of infants are likely to be a vital determinant of postneonatal mortality [5]. It is important to further evaluate this finding in other countries with larger gaps in the quality of and access to postnatal care.



From the public health perspective, the findings from this study improve our understanding of the underlying pathways from maternal education to infant mortality, which should be taken into consideration when designing preventative strategies. The overall educational disparities in infant mortality could possibly be reduced by tackling different intermediates along the pathways. First of all, it is critical to understand and identify socially related prenatal risk factors of PTB and restricted fetal growth. Poor maternal health has been suggested to be strongly associated with adverse birth characteristics [42]. Intervention strategies focused on improving the health of socially and economically disadvantaged women before and during pregnancy to reduce the risk of PTB and SGA may help to prevent neonatal mortality [31,42]. For postneonatal mortality, we found that pathways other than PTB and SGA may play a more critical role. Educational attainment of women is crucial for their access to labor market, income, social resources, financial status, and health behaviors [43]. Education level may have a direct impact on infant health through housing conditions, occupational status, lifestyle, diet, psychosocial stress, compliance with medical advice, the quality of medical care, and the use of specialized medical care [44]. Therefore, efforts to reduce postneonatal deaths need to minimize the link between maternal education and such factors [45]. Reducing these disparities in infant care will require a significant and coordinated effort from different sectors, such as health, housing, labor, and education. Even in a welfare society such as Denmark, pregnant women with low education need more attention and resources to address socially related risk factors to improve infant health in general, and reduce infant mortality.

Strengths and limitations

The unique methodological strength of this study is that we have sufficient and good quality data for performing modern causal mediation analysis in a large study population. Causal mediation analysis can help to explain the mechanisms behind the impact of maternal socioeconomic disadvantage on infant mortality and inform the process of designing public health interventions that will prevent infant deaths. Our study estimated the joint mediating effect of PTB and SGA using the weighting-based approach for causal mediation analysis, which does not necessarily require knowing the ordering of the mediators and is able to address the possibility that the mediators affect each other. It allowed for possible mediator–exposure and mediator–mediator interactions. Furthermore, our large population-based cohort study with almost complete follow-up provided high-quality prospective data and minimized the potential influence of selection bias and recall bias.

The results of the study should be carefully interpreted given the following limitations. First, for valid inferences, causal mediation analysis requires the untestable assumption of there being no unmeasured confounders of the association between infant mortality, maternal education, and the mediators [23,28,33]. Although we adjusted for a wide range of confounders, we cannot exclude residual confounding by unmeasured maternal demographic and lifestyle factors, such as alcohol consumption, body mass index, physical inactivity, psychological stress in adolescence, or school attendance. We applied sensitivity analysis [23,29] to assess the potential impact of unmeasured mediator–outcome confounders on the main results. The results from sensitivity analysis indicated that fairly substantial confounding would be required to explain away our reported results. In addition, some of these unmeasured variables can also be considered as mediators in the pathway between maternal education and infant mortality, in which case they should not be included in the model. Second, ethnicity could be a confounder for both exposure—outcome association and exposure—mediator association. In the US and several European countries, ethnic minority groups have consistently lower



birthweight than the predominant ethnic group. Ethnic disparity is likely due to variations in access to antenatal care and socio-environmental factors, such as workload, stress, and diet [46–48]. However, ethnic disparity is unlikely to be a major concern in Denmark as 91.8% of Danish women are of Danish descent [49], and analyses with additional control for country of origin of the mother (>90% of mothers originally from Denmark [50]) did not change the results. Third, average maternal education levels have increased between 1981 and 2015 in Denmark. The women with only primary and lower secondary education may have become a highly selected group at the end of the period due to increased average education level. Therefore, the stronger association between maternal education and infant mortality observed in recent years could be due to a higher prevalence of risk factors for infant mortality among women with low education. In addition, the same educational attainment does not necessarily reflect the same classifying functions in different calendar periods. The improved educational attainment over the past decades may influence the associations observed in the study. However, mediation analyses stratified by birth year interval found that the results in different intervals were similar to those from the main analyses using data during the whole study period. Finally, our study is based on register data from Denmark, which has universal health coverage for all essential healthcare services. It is important to evaluate these findings in countries with larger disparities in terms of access and the quality of perinatal healthcare.

Conclusions

The association between low maternal education and infant mortality mediated through PTB and SGA was large for neonatal mortality but small for postneonatal mortality. Public health preventive strategies for education-related neonatal mortality in high-income countries may need to address the socially related prenatal risk factors of PTB and impaired fetal growth. On the other hand, the substantial direct association of maternal education with deaths not accounted for by PTB and SGA during the postneonatal period could reflect unaddressed educational disparities in infant care or other factors.

Supporting information

S1 Checklist. STROBE checklist for reporting cohort studies. (DOCX)

S1 Fig. The contribution of preterm birth and small for gestational age in explaining the association between maternal education and infant mortality in Denmark.

S1 Table. The associations between maternal education, preterm birth, small for gestational age, and infant mortality.

(DOCX)

S2 Table. The joint contribution of preterm birth and small for gestational age in explaining the association between maternal education and infant mortality according to birth year.

(DOCX)

S3 Table. The individual contribution of preterm birth and small for gestational age in explaining the association between maternal education and infant mortality according to birth year.

(DOCX)



S4 Table. The contribution of preterm birth in explaining the association between maternal education and mortality among infants that are not small for gestational age. (DOCX)

S5 Table. The contribution of small for gestational age in explaining the association between maternal education and mortality among term-born infants. (DOCX)

S6 Table. The contribution of preterm birth and small for gestational age in explaining the association between maternal education and infant mortality due to certain conditions originating in the perinatal period.

(DOCX)

S7 Table. The contribution of preterm birth and small for gestational age in explaining the association between maternal education and infant mortality due to congenital malformations.

(DOCX)

S8 Table. The contribution of preterm birth and small for gestational age in explaining the association between maternal education and infant mortality due to external causes. (DOCX)

S9 Table. Sensitivity analyses of the influence of unmeasured mediator-outcome confounding.

(DOCX)

S10 Table. The contribution of preterm birth and small for gestational age in explaining the association between maternal education and infant mortality with an additional adjustment for maternal country of origin.

(DOCX)

S11 Table. The contribution of preterm birth and small for gestational age in explaining the association between maternal education and infant mortality using the traditional approach with and without mediators for mediation analysis.

(DOCX)

S1 Text. Analysis plan.

(DOCX)

S2 Text. A detailed description of the inverse probability weighting approach for causal mediation analysis.

(DOCX)

Author Contributions

Conceptualization: Yongfu Yu, Jørn Olsen, Bo Fu, Jiong Li.

Data curation: Yongfu Yu, Carsten Obel, Jiong Li.

Formal analysis: Yongfu Yu, Zeyan Liew, Aolin Wang, Onyebuchi A. Arah, Sven Cnattingius, Guoyou Qin, Carsten Obel, Bo Fu, Jiong Li.

Funding acquisition: Yongfu Yu, Jiong Li.

Investigation: Yongfu Yu, Jørn Olsen, Jiong Li.



Methodology: Yongfu Yu, Zeyan Liew, Aolin Wang, Onyebuchi A. Arah, Jialiang Li, Jørn Olsen, Sven Cnattingius, Guoyou Qin, Bo Fu, Jiong Li.

Project administration: Yongfu Yu, Bo Fu, Jiong Li.

Resources: Yongfu Yu, Bo Fu, Jiong Li.

Software: Yongfu Yu, Aolin Wang, Onyebuchi A. Arah, Bo Fu.

Supervision: Onyebuchi A. Arah, Jørn Olsen, Guoyou Qin, Bo Fu, Jiong Li.

Validation: Yongfu Yu, Bo Fu, Jiong Li.

Visualization: Yongfu Yu, Zeyan Liew, Aolin Wang, Onyebuchi A. Arah.

Writing - original draft: Yongfu Yu.

Writing – review & editing: Yongfu Yu, Zeyan Liew, Aolin Wang, Onyebuchi A. Arah, Jialiang Li, Jørn Olsen, Sven Cnattingius, Guoyou Qin, Carsten Obel, Bo Fu, Jiong Li.

References

- Viner RM, Hargreaves DS, Coffey C, Patton GC, Wolfe I. Deaths in young people aged 0–24 years in the UK compared with the EU15+ countries, 1970–2008: analysis of the WHO Mortality Database. Lancet. 2014; 384(9946):880–92. https://doi.org/10.1016/S0140-6736(14)60485-2 PMID: 24929452
- Calling S, Li X, Sundquist J, Sundquist K. Socioeconomic inequalities and infant mortality of 46,470 preterm infants born in Sweden between 1992 and 2006. Paediatr Perinat Epidemiol. 2011; 25(4):357–65. https://doi.org/10.1111/j.1365-3016.2011.01200.x PMID: 21649678
- Arntzen A, Samuelsen SO, Daltveit AK, Stoltenberg C. Post-neonatal mortality in Norway 1969–95: a cause-specific analysis. Int J Epidemiol. 2006; 35(4):1083–9. https://doi.org/10.1093/ije/dyl047 PMID: 16556644
- Wood AM, Pasupathy D, Pell JP, Fleming M, Smith GC. Trends in socioeconomic inequalities in risk of sudden infant death syndrome, other causes of infant mortality, and stillbirth in Scotland: population based study. BMJ. 2012; 344:e1552. https://doi.org/10.1136/bmj.e1552 PMID: 22427307
- Arntzen A, Samuelsen SO, Bakketeig LS, Stoltenberg C. Socioeconomic status and risk of infant death. A population-based study of trends in Norway, 1967–1998. Int J Epidemiol. 2004; 33(2):279–88. https://doi.org/10.1093/ije/dyh054 PMID: 15082627
- Strand BH, Groholt EK, Steingrimsdottir OA, Blakely T, Graff-Iversen S, Naess O. Educational inequalities in mortality over four decades in Norway: prospective study of middle aged men and women followed for cause specific mortality, 1960–2000. BMJ. 2010; 340:c654. https://doi.org/10.1136/bmj.c654 PMID: 20179132
- Mackenbach JP, Bos V, Andersen O, Cardano M, Costa G, Harding S, et al. Widening socioeconomic inequalities in mortality in six Western European countries. Int J Epidemiol. 2003; 32(5):830–7. https:// doi.org/10.1093/ije/dyg209 PMID: 14559760
- 8. Heron M. Deaths: leading causes for 2010. Natl Vital Stat Rep. 2013; 62(6):1–96. PMID: 24364902
- Weightman AL, Morgan HE, Shepherd MA, Kitcher H, Roberts C, Dunstan FD. Social inequality and infant health in the UK: systematic review and meta-analyses. BMJ Open. 2012; 2(3):e000964. https://doi.org/10.1136/bmjopen-2012-000964 PMID: 22700833
- Spencer N. The effect of income inequality and macro-level social policy on infant mortality and low birthweight in developed countries—a preliminary systematic review. Child Care Health Dev. 2004; 30 (6):699–709. https://doi.org/10.1111/j.1365-2214.2004.00485.x PMID: 15527480
- Parker JD, Schoendorf KC, Kiely JL. Associations between measures of socioeconomic status and low birth weight, small for gestational age, and premature delivery in the United States. Ann Epidemiol. 1994; 4(4):271–8. PMID: 7921316
- Katz J, Lee AC, Kozuki N, Lawn JE, Cousens S, Blencowe H, et al. Mortality risk in preterm and smallfor-gestational-age infants in low-income and middle-income countries: a pooled country analysis. Lancet. 2013; 382(9890):417–25. https://doi.org/10.1016/S0140-6736(13)60993-9 PMID: 23746775
- Ruiz M, Goldblatt P, Morrison J, Kukla L, Svancara J, Riitta-Jarvelin M, et al. Mother's education and the risk of preterm and small for gestational age birth: a DRIVERS meta-analysis of 12 European cohorts. J Epidemiol Community Health. 2015; 69(9):826–33. https://doi.org/10.1136/jech-2014-205387 PMID: 25911693



- Mortensen LH, Lauridsen JT, Diderichsen F, Kaplan GA, Gissler M, Andersen AM. Income-related and educational inequality in small-for-gestational age and preterm birth in Denmark and Finland 1987– 2003. Scand J Public Health. 2010; 38(1):40–5. https://doi.org/10.1177/1403494809353820 PMID: 19906773
- Leon DA. Influence of birth weight on differences in infant mortality by social class and legitimacy. BMJ. 1991; 303(6808):964–7. https://doi.org/10.1136/bmj.303.6808.964 PMID: 1954421
- Suhag A, Berghella V. Intrauterine growth restriction (IUGR): etiology and diagnosis. Curr Obstet Gynecol Rep. 2013; 2(2):102–11. https://doi.org/10.1007/s13669-013-0041-z
- Gardosi JO. Prematurity and fetal growth restriction. Early Hum Dev. 2005; 81(1):43–9. https://doi.org/ 10.1016/j.earlhumdev.2004.10.015 PMID: 15707714
- Ray JG, Park AL, Fell DB. Mortality in infants affected by preterm birth and severe small-for-gestational age birth weight. Pediatrics. 2017; 140(6): e20171881. https://doi.org/10.1542/peds.2017-1881 PMID: 29117948
- Li J, Vestergaard M, Obel C, Cnattingus S, Gissler M, Olsen J. Cohort profile: the Nordic Perinatal Bereavement Cohort. Int J Epidemiol. 2011; 40(5):1161–7. https://doi.org/10.1093/ije/dyq127 PMID: 20675718
- Gissler M, Mohangoo AD, Blondel B, Chalmers J, Macfarlane A, Gaizauskiene A, et al. Perinatal health monitoring in Europe: results from the EURO-PERISTAT project. Inform Health Soc Care. 2010; 35 (2):64–79. https://doi.org/10.3109/17538157.2010.492923 PMID: 20726736
- Eurostat. European shortlist for causes of death, 1998. Luxembourg: Eurostat; 2015 [cited 2015 Oct 17]. Available from: http://ec.europa.eu/eurostat/ramon/documents/cod_1998/cod_1998_pdf.zip.
- Bliddal M, Broe A, Pottegard A, Olsen J, Langhoff-Roos J. The Danish Medical Birth Register. Eur J Epidemiol. 2018; 33(1):27–36. https://doi.org/10.1007/s10654-018-0356-1 PMID: 29349587
- VanderWeele T. Explanation in causal inference: methods for mediation and interaction. Oxford: Oxford University Press; 2015.
- Pearl J. Direct and indirect effects. In: Breese J, Koller D, editors. Proceedings of the seventeenth conference on uncertainty in artificial intelligence. San Francisco: Morgan Kaufmann Publishers; 2001. pp. 411–20.
- Robins JM, Greenland S. Identifiability and exchangeability for direct and indirect effects. Epidemiology. 1992; 3(2):143–55. PMID: 1576220
- **26.** Robins JM, Hernan MA, Brumback B. Marginal structural models and causal inference in epidemiology. Epidemiology. 2000; 11(5):550–60. PMID: 10955408
- VanderWeele TJ. Policy-relevant proportions for direct effects. Epidemiology. 2013; 24(1):175–6. https://doi.org/10.1097/EDE.0b013e3182781410 PMID: 23232624
- 28. VanderWeele TJ. Mediation analysis: a practitioner's guide. Annu Rev Public Health. 2016; 37:17–32. https://doi.org/10.1146/annurev-publhealth-032315-021402 PMID: 26653405
- 29. VanderWeele TJ. Bias formulas for sensitivity analysis for direct and indirect effects. Epidemiology. 2010; 21(4):540–51. https://doi.org/10.1097/EDE.0b013e3181df191c PMID: 20479643
- Strandberg-Larsen K, Gronboek M, Andersen AM, Andersen PK, Olsen J. Alcohol drinking pattern during pregnancy and risk of infant mortality. Epidemiology. 2009; 20(6):884–91. https://doi.org/10.1097/EDE.0b013e3181bbd46c PMID: 19797967
- Arntzen A, Mortensen L, Schnor O, Cnattingius S, Gissler M, Andersen AM. Neonatal and postneonatal mortality by maternal education—a population-based study of trends in the Nordic countries, 1981– 2000. Eur J Public Health. 2008; 18(3):245–51. https://doi.org/10.1093/eurpub/ckm125 PMID: 18160387
- Guildea ZE, Fone DL, Dunstan FD, Sibert JR, Cartlidge PH. Social deprivation and the causes of still-birth and infant mortality. Arch Dis Child. 2001; 84(4):307–10. https://doi.org/10.1136/adc.84.4.307
 PMID: 11259227
- **33.** Richiardi L, Bellocco R, Zugna D. Mediation analysis in epidemiology: methods, interpretation and bias. Int J Epidemiol. 2013; 42(5):1511–9. https://doi.org/10.1093/ije/dvt127 PMID: 24019424
- Spencer N, Bambang S, Logan S, Gill L. Socioeconomic status and birth weight: comparison of an area-based measure with the Registrar General's social class. J Epidemiol Community Health. 1999; 53(8):495–8. https://doi.org/10.1136/jech.53.8.495 PMID: 10562868
- Valero De Bernabe J, Soriano T, Albaladejo R, Juarranz M, Calle ME, Martinez D, et al. Risk factors for low birth weight: a review. Eur J Obstet Gynecol Reprod Biol. 2004; 116(1):3–15. https://doi.org/10. 1016/j.ejogrb.2004.03.007 PMID: 15294360
- **36.** Olsen O, Madsen M. Effects of maternal education on infant mortality and stillbirths in Denmark. Scand J Public Health. 1999; 27(2):128–36. PMID: 10421722



- Arntzen A, Nybo Andersen AM. Social determinants for infant mortality in the Nordic countries, 1980–2001. Scand J Public Health. 2004; 32(5):381–9. https://doi.org/10.1080/14034940410029450 PMID: 15513672
- Yu Y, Qin G, Cnattingius S, Gissler M, Olsen J, Zhao N, et al. Mortality in children aged 0–9 years: a nationwide cohort study from three Nordic countries. PLoS ONE. 2016; 11(1):e0146669. https://doi.org/10.1371/journal.pone.0146669 PMID: 26744840
- Wilcox AJ. On the importance—and the unimportance—of birthweight. Int J Epidemiol. 2001; 30 (6):1233–41. https://doi.org/10.1093/ije/30.6.1233 PMID: 11821313
- **40.** EURO-PERISTAT project, Surveillance of Cerebral Palsy in Europe, EUROCAT, EURONEOCAT. European perinatal health report. EURO-PERISTAT; 2008 [cited 2015 May 12]. Available from: https://www.europeristat.com/images/doc/EPHR/european-perinatal-health-report.pdf.
- Halldorsson M. Socioeconomic differences in children's use of physician services in the Nordic countries. J Epidemiol Community Health. 2002; 56(3):200–4. https://doi.org/10.1136/jech.56.3.200 PMID: 11854341
- Zylbersztejn A, Gilbert R, Hjern A, Wijlaars L, Hardelid P. Child mortality in England compared with Sweden: a birth cohort study. Lancet. 2018; 391(10134):2008–18. https://doi.org/10.1016/S0140-6736(18)30670-6 PMID: 29731173
- 43. Garcia-Altes A, Ruiz-Munoz D, Colls C, Mias M, Martin Bassols N. Socioeconomic inequalities in health and the use of healthcare services in Catalonia: analysis of the individual data of 7.5 million residents. J Epidemiol Community Health. 2018; 72(10):871–9. https://doi.org/10.1136/jech-2018-210817 PMID: 30082426
- 44. Renard F, Gadeyne S, Devleesschauwer B, Tafforeau J, Deboosere P. Trends in educational inequalities in premature mortality in Belgium between the 1990s and the 2000s: the contribution of specific causes of deaths. J Epidemiol Community Health. 2017; 71(4):371–80. https://doi.org/10.1136/jech-2016-208370 PMID: 27885048
- 45. Chen J, Fair M, Wilkins R, Cyr M, Fetal and Infant Mortality Study Group of the Canadian Perinatal Surveillance System. Maternal education and fetal and infant mortality in Quebec. Health Rep. 1998; 10 (2):53–64. PMID: 9842491
- 46. Kramer MS, Seguin L, Lydon J, Goulet L. Socio-economic disparities in pregnancy outcome: why do the poor fare so poorly? Paediatr Perinat Epidemiol. 2000; 14(3):194–210. https://doi.org/10.1046/j.1365-3016.2000.00266.x PMID: 10949211
- Villadsen SF, Mortensen LH, Andersen AM. Ethnic disparity in stillbirth and infant mortality in Denmark 1981–2003. J Epidemiol Community Health. 2009; 63(2):106–12. https://doi.org/10.1136/jech.2008.078741 PMID: 18930979
- 48. Kelly Y, Panico L, Bartley M, Marmot M, Nazroo J, Sacker A. Why does birthweight vary among ethnic groups in the UK? Findings from the Millennium Cohort Study. J Public Health (Oxf). 2009; 31(1):131–7. https://doi.org/10.1093/pubmed/fdn057 PMID: 18647751
- Statistics Denmark. Population and elections. Copenhagen: Statistics Denmark; 2017 [cited 2017 Sep 28]. Available from: http://www.statbank.dk/statbank5a/default.asp?w=1920.
- Plana-Ripoll O, Liu X, Momen NC, Parner E, Olsen J, Li J. Prenatal exposure to maternal stress following bereavement and cardiovascular disease: a nationwide population-based and sibling-matched cohort study. Eur J Prev Cardiol. 2016; 23(10):1018–28. https://doi.org/10.1177/2047487315585294
 PMID: 25952251